Author’s response to reviews

Title: ATM rs189037 (G>A) polymorphism increased the risk of cancer: an updated meta-analysis

Authors:

Zhi-liang Zhao (zzl81@hotmail.com)

Lu Xia (492193335@qq.com)

Cong Zhao (17036251@qq.com)

Jun Yao (yaojun198717@163.com)

Version: 2 Date: 18 Jan 2019

Author’s response to reviews:

Dear Editor,

Thanks so much for your work on my submission. We have revised the manuscript in light of your comments, as follows:

Comment 1. We note that the current submission contains some textual overlap with other previously published works (37%), in particular:

12% - "No association between dopamine D3 receptor gene Ser9Gly polymorphism (rs6280) and risk of schizophrenia: an updated meta-analysis" (2017) Neuropsychiatric Disease and Treatment DOI: 10.2147/NDT.S152784

and

4% - "Association between DRD2 (rs1799732 and rs1801028) and ANKK1 (rs1800497) polymorphisms and schizophrenia: A meta-analysis" (2014) American Journal of Medical Genetics DOI: 10.1002/ajmg.b.32281

This overlap exists throughout the manuscript. While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work. Please re-phrase these sections to minimise overlap. Please see the attached screenshot of the overlap.
Reply. We have rewritten the overlap description in the revised manuscript.

Comment 2. Please amend the Abstract headings to: Background, Methods, Results, Conclusions.

Reply. We have modified the Abstract according to your suggestion.

Comment 3. Please amend the main text headings to: Background, Methods, Results, Discussion, Conclusions.

Reply. We have modified the main text headings according to your suggestion.

Comment 4. Please represent authors' names using their initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Reply. We have represented authors' names using their initials in the Authors’ Contributions section.

Comment 5. Please remove the response to reviewers' comments letter, as it is no longer needed at this stage. Please do not upload a new letter with this revision.

Reply. We have removed the response to reviewers' comments letter.

Comment 6. Please remove the PRISMA checklist, as it is no longer needed at this stage.

Reply. We have removed the PRISMA checklist.

Comment 7. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Reply. We have uploaded the manuscript according to your comment.